ORIGINAL ARTICLE



Friend-or-foe? Behavioural evidence suggests interspecific discrimination leading to low probability of hybridization in two coexisting rock lizard species (Lacertidae, *Darevskia*)

Eduard A. Galoyan^{1,2} · Elena Yu. Tsellarius² · Marine S. Arakelyan³

Received: 23 October 2018 / Revised: 19 February 2019 / Accepted: 21 February 2019 © Springer-Verlag GmbH Germany, part of Springer Nature 2019

Abstract

The parthenogenetic rock lizard *Darevskia rostombekowi* is considered to be the product of hybridization between female *Darevskia raddei* and male *Darevskia portschinskii*. These two species coexist within several secondary contact areas; however, no trace of their hybridization has been previously reported. We conducted focal observations of individually marked lizards in 2017 and 2018 to establish if there is behavioural isolation between these species. We demonstrate that individuals distinguish between lizards of the same and different species. Individuals of both studied species interacted regularly, but we found no evidence of males of either species using interspecific interference competition for resources or for females. Neutral reactions prevailed in the reactions of males to the individuals of the different species; aggressive or submissive behaviour was more common towards males of the same species. The differences in reactions of males to females of the same and different species were almost as common as intraspecific interactions. Interspecific male mating behaviour was rare and unsuccessful; only a single female with heterospecific copulation marks on the body was found. The mating period in both species overlapped broadly in late May and early June; body size (SVL) in females of both species was equal. Hence, we suggest that the selection of the sexual partner and preference of the conspecific partner for the social and sexual contacts is the most plausible explanation of the revealed behavioural asymmetry in intra- and interspecific relationship.

Significance statement

Here, we present the first description of the social and sexual behaviour of two bisexual species of rock lizard belonging to the *Darevskia* genus. Most works devoted to the study of reticulate evolution in lizards have used genetic and morphological approaches rather than behavioural observations. The study of the proximal mechanisms of evolution is important in understanding how evolution occurs. The results indicate that even in ecologically similar species, social and sexual selection tend towards separation of the species rather than mixing, although they are able to coexist peacefully.

Keywords Social behaviour · Hybridization · Darevskia · Reproduction barrier · Species recognition

Communicated by S. J. Downes

Electronic supplementary material The online version of this article (https://doi.org/10.1007/s00265-019-2650-7) contains supplementary material, which is available to authorized users.

Eduard A. Galoyan saxicola@mail.ru

- ¹ Zoological Museum, Lomonosov Moscow State University, Moscow 125009, Russia
- ² A. N. Severtsov Institute of Ecology and Evolution, Moscow 119071, Russia
- ³ Department of Zoology, Yerevan State University, 0025 Yerevan, Armenia

Introduction

Of the more than 30 species of rock lizard (*Darevskia* spp.), seven are parthenogenetic. Their origin and role in reticulate evolution of the complex have been studied mostly from a genetic and cytogenetic point of view (Cuellar 1974; Borkin and Darevsky 1980; Vrijenhoek 1989; Cole and Townsend 1990; Darevsky 1995; Grismer and Grismer 2010; Freitas et al. 2016; Spangenberg et al. 2017). As a result, there is an extensive body of data supporting the hypothesis of origin of each parthenogenetic species from hybridization between different bisexual taxa of the same genus (Darevsky et al. 1985; Cole et al. 1988; Moritz et al.

1992; Fu et al. 2000a, b; Murphy et al. 2000; Grismer et al. 2014).

The possibility of hybridization between different bisexual species is supported under laboratory conditions (Danielyan 1965; Darevsky et al. 1985; Darevsky 1995). Although in some experiments only sterile hybrids of the first generations were bred (Vrijenhoek 1989), other experiments had more promising results. The first artificial breeding of a unisexual fish belonging to the genus Poeciliopsis was reported by Schultz (1973), and the first parthenogenetic lizard Aspedocelis (Teidae) was obtained fairly recently by Cole et al. (2014). These experiments support the potential of interspecific hybridization to lead to the origin of new species of vertebrates, including parthenogenetic ones. However, there is a lack of both genetic and experimental data on the relationships and interactions of heterospecific individuals to each other in natural habitats. Moreover, the proximal mechanisms resulting in hybridization between different related species remain poorly studied (Wirtz 1999; Rosenthal 2013).

Interspecific mating and hybridization generally occur within the areas of secondary species contacts. This is supported by studies of *Aspedocelis* (Dessauer et al. 2000), *Podarcis* (Pinho et al. 2009), *Vipera* (Tarroso et al. 2014) and other reptiles (Axtell 1972; Jenssen 1977; Jančúchová-Lásková et al. 2015; Panov and Zykova 2016). In some cases of secondary contacts, gene flow between species is absent (Gabirot et al. 2010). Likewise, parthenogenetic lizards often coexist and mate with the males of parental species (Darevsky et al. 1985; Taylor et al. 2001; Danielyan et al. 2008; Jančúchová-Lásková et al. 2015). Hence, range overlap is a key precondition for hybridization.

Two types of syntopy between bisexual species of rock lizard are described by Darevsky (1958). For the first type, the area of distribution of one species overlaps the area of another species. For instance, a small number of *D. valentini* (paternal species of *D. unisexualis*) individuals have been found within the range of *D. raddei nairensis* (maternal species) in the area of Lake Sevan, Armenia (Danielyan 1965; Darevsky et al. 1985). However, no hybrids were recorded there, most likely due to different timing of reproduction and gonadal ripening (Danielayn 1965). The second type of species coexistence is characterised by almost equal numbers of individuals of both syntopic species (Arakelyan 2012).

Darevskia raddei and *D. portschinskii* are proposed to be parental for *D. rostombekowi* (Fu et al. 2000b; Murphy et al. 2000; Ryskov et al. 2017). To date, the ranges of these parental species are mostly allopatric. One contact zone between them has been found in the vicinity of the village of Gosh in Armenia, where about 9% of individuals of *D. portschinskii* occur within the range of *D. raddei* (Arakelyan and Danielyan 2014). However, a study of allozymes in *D. raddei*, *D. portschinskii*, and plausible hybrids between them revealed no evidence of hybridization in this area (Uzzel and Darevsky 1973). In contrast to the population in Gosh, a large zone of syntopy, representing this second type of hybrid zone, was found for *D. portschinskii* and *D. raddei* in 2004 in the vicinity of Zuar village in Artsakh (Arakelyan 2012). Again, no hybrids have been discovered here (Galoyan et al. 2019).

According to the hypothesis of hybridogenous speciation, the parental species for D. rostombekowi (D. raddei and D. porschinskii) should lack interspecific reproductive barrier. However, their stable coexistence within the secondary contact zones and the absence of any evidence of hybridization suggests the existence of such a barrier. Similar bimodal hybrid zones with some, but not absolute, isolation have been reported for Podarcis (Pinho et al. 2009). Our goal was to determine if there are any behavioural mechanisms of reproductive isolation between D. raddei and D. portschinskii in the area of their coexistence. We suggest that the lizards' reactions to individuals of the same and different species must be different if they can distinguish between species and if some prezygotic reproduction barriers are present. Otherwise, the males of both species should mate with females of both species and compete for them with males of both species. Therefore, our aim was to analyse the type and frequency of contacts between two coexisting species of rock lizard, which are parental to D. rostombekowi in the area of their sympatry.

Methods

Study species

According to Arribas (1999), *D. portschinskii* (Kessler 1878) belongs to the 'rudis' clade, while *D. raddei* (Boettger 1892) is part of the 'raddei' clade within the *Darevskia* genus. Both species are diurnal lizards distributed to the south of the Great Caucasus, and are found up to an elevation of 1500 m.a.s.l. for *D. portschinskii* and 3200 m.a.s.l. for *D. raddei*.

Both studied species have a slim and relatively small body, with a flattened head well-adapted to living among rocks and squeezing within narrow crevices. SVL of D. portschinskii reaches 67 mm and SVL of D. raddei reaches 69 mm (Darevsky 1967; Arakelyan et al. 2011). Males of D. raddei are larger than females and have similar body size to D. portschinskii (Darevsky 1967). Body proportions are different between sexes for both species: males have wider and more robust heads and females have longer bodies (Arakelyan et al. 2011; Dehghani et al. 2014). Dorsum coloration is brown (greenish in adult males) with dark indistinct blotches and two wide flank bands in D. portschinskii; brown with reticulate dark pattern in D. raddei. Belly is light or bright yellow for D. portschinskii and green-yellow for D. raddei. Most species of rock lizards, including D. raddei and D. portschinskii, become sexually mature adults after their third winter, when their SVL exceed 50 mm (Darevsky 1967; Arakelyan 2001).

In general, *D. raddei* prefers dryer and sunlit habitats, while *D. portschinskii* inhabits rocky habitats along rivers and in deciduous forests (Arakelyan et al. 2011). In some areas, both species tend to occupy the same rocky steppes (Uzzel and Darevsky 1973).

Study area

The survey area was located near the village of Zuar in the south-eastern most part of the Lesser Caucasus Mountains at a height of 1500 m.a.s.l. on the rocky southern slopes, comprising the easternmost part of the range for both *D. portschinskii* and *D. raddei*. Here, deep river valleys lie between watersheds, where rock lizards occupy the rocky outcrops along the roads and trails within the gorges, as well as the rocks and stony heaps on the slopes and the edges of the watersheds covered with deciduous oak forests and grass meadows.

A rocky plot of about 4000 m^2 in the area of the village was chosen as the survey area. A 2.5-m high stony wall and the remains of an old abandoned building were within the study site. There was also a meadow with a few trees (birch, medlar, plum and hornbeam). Here, the lizards mostly occupied the walls, using the cracks between stones as shelter, as well as wall debris and stony heaps, although they also moved from one rocky patch to another crossing the vast areas of grassland.

Field work

The field survey covered the period between 23 May to 27 June 2017 (27 working days) and 9 May to 21 July 2018 (33 working days). According to our observations, this timing covers the reproduction period of both species in the studied area. Mating period in D. portschinskii starts in mid-May and ends in early June, and it starts in late May and ends in late June in D. raddei (ESM: Table S1). Adult lizards (snout vent length [SVL] > 50 mm) were captured within the study area by hand or by noose. Each lizard was individually marked using a combination of two colour beads sewn through the skin fold on the dorsum or flank caudal area using nylon line (Fisher and Muth 1989). This method makes it possible to observe and distinguish individuals from a distance. In addition, photographic identification was used to identify individuals. The described marking technique has no negative impact on the lizard's behaviour (Tsellarius and Tsellarius 2001; Husak and Fox 2003; Nicholson and Richards 2011; Galoyan 2013, 2017).

Most lizards were marked in the first days after emergence from their winter shelters and newcomers were marked as we encountered them. We measured the SVL of each individual using callipers with an accuracy of 0.5 mm and photographed the dorsal pattern of the lizard for future identification.

Visual observations

Visual observations of individually marked lizards were performed between 08:00 and 18:00 h on days with favourable weather conditions, when the lizards were active. Focal observations were made by one or two observers from 5 to 10 m or closer, when the lizards had become habituated to the presence of the observer. It was not possible to record data blind because our study involved focal animals in the field. We moved slowly to minimise any disturbance to the lizards. After several days, observed individuals paid less attention to the observer (did not switch from routine behaviour when the observer approached or could pass near the observer). The observer chose a focal lizard and observed the individual for 10-15 min, before switching to another individual. The location and contacts, including successful or unsuccessful mating attempts, and foraging behaviour, were registered and recorded using a Nikon D600 camera whenever possible, or were written down in the field book. All records and videos were processed at the end of each observation day, by the same person. The total observation time was 141 h in 2017 and 164 h in 2018.

Interactions and relationship

Individual ranges overlapped broadly among the observed individuals of both species and lizards had the opportunity to interact with individuals of both species. We considered an interaction to be an encounter between two or more individuals within the reaction distance, which is less than or around 1 m for rock lizards (Galoyan 2013), if the individuals are clearly able to see each other (there are no rocks or logs between potential contactors). In total, we observed 227 contacts in 2 years (Table S2). We did not have enough data for contacts between females for analysis. Interspecific contacts were as common as intraspecies contacts (Table 1). We constructed a social network based on all interactions observed in 2017 using the 'igraph' and 'network' packages in R (Csardi and Nepusz 2006).

Each contact started with a direct reaction, e.g. (the reaction of individual 'A' to individual 'B' before B switched from routine (moving, foraging, etc.) to social behaviour.

		Males	
		D. raddei	D. portschinskii
Male	D. raddei	44 (37.3%)	55 (46.6%)
	D. portschinskii		19 (16.1%)
Female	D. raddei	48 (46.2%)	10 (9.6%)
	D. portschinskii	14 (13.5%)	32 (30.7%)

Reciprocal reaction was generated in response to the direct reaction (Tsellarius et al. 2008). We also examined neutral interactions, which occurred if the lizards approached each other to the reaction distance and could see each other but did not change their behaviour. Males often initiate intersexual interactions (Verbeek 1972), whereas females mostly react as a response to the reaction of the particular male. Hence, when analysing intersexual contacts, we only took into account male reactions to the females and excluded females' reactions. We also only used interactions in which both interactors were identified and marked.

The following types of social reaction during the interactions between individuals were described: neutral, agonistic (aggressive and submissive), orientation and friendly behaviour. Neutral behaviour was defined as the absence of any reaction to the conspecific (no change in behaviour, Video S1). Agonistic behaviour included aggression and submission. Aggression was characterised by 'threat' with the individual extending its limbs, curving the dorsum, flattening the body flanks and gular pouch and moving with short steps often around the encounter; 'attack' involved chasing and biting the second contactor (Video S1, S2). Submissive behaviour comprised 'submission', represented by circumduction (slow or fast forelimb rotation), often with slow retreat from the encounter area before actual contact with the partner happened, or in the case of a neutral reaction from the partner; 'flee' entailed leaving the encounter area promptly. Orientation was characterised by 'approach to the partner', often with tongue licking. The approaching individual could also push the partner with the tip of the head. Friendly behaviour was characterised by jointly basking in a physical contact or in one or two centimetres from each other. Individuals could also overlap each other: one lizard gets on the back of the other and then goes further; we termed this a friendly reaction (Video S3).

Sexual behaviour included courtship and mating attempts, which could differ depending on the relationship between the particular individuals (Tsellarius et al. 2016). Sexual harassment and mating attempts included chasing and forced restraint of a female. Usually, females tried to escape and could respond aggressively to the male's harassment (biting the male, Video S4). Non-aggressive mating was often preceded by joint basking with physical contact (friendly behaviour). We termed a mating attempt successful if it ended with copulation and unsuccessful if not.

Statistical analysis

The distribution of the body lengths (SVL) was tested for normality using the Shapiro-Wilk normality test. Medians of SVL are reported with minimum and maximum measurements for the lizards from the studied settlement within each species and sex; the comparison among individuals in different years and species was performed with unpaired Wilcoxon test.

We developed three data sets for the reactions of each individual from the pool of observed reactions. Each of the data sets included only one reaction of a focal individual to another individual belonging to one of four categories (male of the same species, male of the different species, female of the same species and female of the different species). Each data set included randomly selected reactions for each focal individual to the individual from the particular category. For instance, if we had three reactions of male A to the females of the same species, we three times randomly selected one of these reactions using a random selection command in the R package. This method allowed us to avoid the effect of a bias of contacts for particular individuals. Then, we compared the distribution of reaction frequencies among the datasets using the chisquare (χ^2) test to reveal the absence of differences in the distributions of reactions. The final dataset for analysis included medians of reactions for the three datasets (Table S2). We calculated the proportion of the reactions of males of both species to the males and females of both species and compared the distribution of reactions using unpaired Wilcoxon test.

The average copulation time was estimated based on fully observed copulations (N=9) and given with the standard error (SE). The Excel of the 2007 Microsoft Office was used for making the database and 'ggplot2' 3.1.0. data visualisation package in R for making plots.

Data availability

All data generated or analysed during this study are included in this published article and its supplementary information files.

Results

Settlement composition

We marked 56 *D. raddei* and 40 *D. portschinskii* (Table S3) in 2017 and 2018. The number of adult and mature animals captured and marked in 2017 comprised 35 *D. raddei* (12 females, 23 males) and 28 *D. portschinskii* (13 females, 15 males). The number of individuals marked in 2018 comprised 37 *D. raddei* (9 recaptured and 9 new males; 9 recaptured and 6 new females) and 28 *D. portschinskii* (11 recaptured, 1 registered, but not recaptured and 7 new males; 8 recaptured and 2 new females). Hence, in both species, there were fewer females than males.

We found no significant differences in body length (SVL) of adult lizards belonging to the same species and sex between years (unpaired Wilcoxon test, p > 0.05). The median body length (SVL) of adult females for *D. raddei* was 55 mm

(min = 51, max = 64 mm, N = 27) and 55 mm (min = 50, max = 60 mm, N = 23) for *D. portschinskii*. We found no significant difference in SVL between adult females of the studied species were revealed (unpaired Wilcoxon test, p > 0.05, Fig. 1, Table S4). The median SVL of adult males of *D. portschinskii* was 54 mm (min = 50, max = 59 mm, N = 33). We found no significant differences between adult male and female body length of the same species (unpaired Wilcoxon test, p > 0.05); however, SVL of adult males of *D. portschinskii* was significantly smaller than this in adult females of *D. raddei* (unpaired Wilcoxon test, W = 254.5, p < 0.01, Fig. 1, Table S4). The largest body size was in males of *D. raddei* (median = 58 mm, min = 50, max = 63 mm, N = 41) and it was significantly larger than in males of *D. portschinskii* (unpaired Wilcoxon test, W = 253, p < 0.01).

Relationships between individuals

In total, we observed 118 M–M, 104 M–F and only 5 F–F interactions (Table 1). Lizards of both species shared space, and the social network depicts regular interactions between individuals of both species (Fig. 2).

Male *D. raddei* reacted differently to the males of the same and different species (unpaired Wilcoxon test, W = 10, p < 0.01, N = 18 and 11, Fig. 3a). The reactions of male *D. portschinskii* to conspecific and heterospecific males also differed significantly (unpaired Wicoxon test, W = 88, p < 0.01, N = 8 and 11, Fig. 3b). The same reactions of the same individuals towards conspecific and heterospecific males confirm it (*D. portschinskii*: Wilcoxon signed-rank test, V = 0.0, N = 7, p = 0.02; *D. raddei*: Wilcoxon signed-rank test,



Fig. 1 Body size (SVL) of adult males (M) and females (F) of *D. raddei* (N = 68) and *D. portschinskii* (N = 56), marked during the study period in 2017 and 2018 years. The significant differences are marked with *



Fig. 2 Interspecific interactions (red lines); interactions between *D. raddei* (green lines) and *D. portshinskii* (blue lines) among adult and subadult males (circles) and females (squares) of *D. raddei* (blue) and *D. portschinskii* (green) in 2017 and 2018 years. $N_{\text{interactions}} = 228$; $N_{D.raddei} = 14$ females and 24 males; $N_{D.portschinskii} = 12$ females and 15 males

V = 55.0, N = 10, p = 0.005). Neutral reactions prevailed when males of two different species met and even rare friendly reactions with physical contact (joint basking) were observed (Fig. 3). The reactions of males of the same species were mostly aggressive or submissive (flee or contact avoidance).

In general, male D. raddei seemed to be more neutral to the females of D. portschinskii than to females of the same species, although unsuccessful mating attempts with females of other species were also recorded (Fig. 4a). The differences in reactions of D. raddei males to females of both species were confirmed statistically (unpaired Wilcoxon test, W = 112, N =16 and 10, p < 0.01), but not in the paired reaction of the same individuals (Wilcoxon signed-rank test, V = 22.0, N = 7, p =0.19). We found no significant differences in the reactions of D. portschinskii males to females of both species (Wilcoxon signed-rank test, N = 8 and 11, p > 0.05, Fig. 4b). Although we had no statistical support for differences between male reactions to the females of the same and different species, our general impression was that males were searching for contacts with conspecifics and interacted with heterospecifics only if met them by chance.

Mating

Among 23 observed mating attempts, only two were interspecific: we observed two *D. raddei* males chase female *D. portschinskii*. None of these attempts were successful (Table 2). We did not observe any attempts by male *D. portschinskii* to mate with 'wrong' females, although in general they were typically indifferent to female *D. raddei* as well. More than half the recorded intraspecific mating attempts were successful (no differences between *D. portschinskii* and *D. raddei*, chi-square test, $\chi^2 = 0$, df =1, p = 1). Fig. 3 Responses of *D. raddei* males (\mathbf{a} , N = 29) and of *D. portschinskii* males (\mathbf{b} , N = 19) to males of the same and different species. The reactions of the particular category on another category are considered as 100%



On average, mating lasted for $6 \pm 1.38 \text{ min}$ (N = 7, max = 12 min) in *D. raddei*. Only two instances of mating were observed for *D. portschinskii*, both much longer than any in *D. raddei* (26 and 50 min, respectively). The ways in which the males restrained females also differed among species. For *D. portchinskii*, jaw prints were left on the side body flank and for *D. raddei* on the femur. Hence, mating attempts could be distinguished by the position of the jaw prints on a female's body (Fig. 5). We saw no females with jaw prints left by males of the different species within the study area; however, on 13 June 2018, a female of *D. raddei* with femur prints (as from a male of *D. raddei*) and with belly prints (as from a male of *D. portschinskii*) was found 100 m from the 2018 study area.

Discussion

The parthenogenetic rock lizard *D. rostombekowi* is considered to be the progeny of hybrids between *D. raddei* as the maternal species and *D. portschinskii* as the paternal species (Ryskov et al. 2017). If this is true, barriers to reproductive isolation have broken down once. The external morphology and body proportions of both studied species are similar (Fig. 4; Darevsky 1967). This should be a good predictor for niche overlap and use of the same resources (Pianka and Vitt 2003) favouring range overlap and existence of the stable secondary contact zones (Uzzel and Darevsky 1973; Arakelyan et al. 2011). However, no evidence of hybridization between *D. raddei* and *D. portschinskii* has been reported (Uzzel and Darevsky 1973; Galoyan et al. 2019). Hence, range overlap is important, but it is not the only condition for hybridization between the two species.

Male reactions to conspecifics and heterospecifics indicate their ability to distinguish between males of the same and different species. Males are known to be more aggressive than females, and direct competition is expressed due to rather high aggression in males (Ruby 1978). Indeed, interactions between males of the same species were always agonistic (aggressive or submissive) and neutral reactions prevailed during interspecific encounters of males. Only a few aggressive encounters were observed between males of different species. These could have happened if in the case of mistaken identity and/or if a heterospecific male also approached the focal male. Males of different species even basked in physical contact **Fig. 4** Responses of *D. raddei* (**a**) males and of *D. portschinskii* (**b**) males to females of the same and different species. The reactions of the particular category on another category are considered as 100%



with each other. Such a high tolerance of males towards heterospecifics of the same sex suggests an absence of interspecific resource competition. The differences in reactions towards conspecifics and heterospecifics cannot be explained by spatial segregation, which is absent in the studied population: individuals of both species interacted regularly. Relatively low number of heterospecific interactions is best explained by the social preferences.

Male reactions to females of the same and different species also differed, although these differences were less distinct than in intrasexual reactions among males. Male *D. raddei* might have chased female *D. portschinskii* (mating attempt), although it is difficult to interpret this unambiguously—we could not tell if these were cases of mistaken identity or whether they did so deliberately. A single female, which was found with interspecific jaw prints, suggests a mating attempt, but we could not say whether this was a successful copulation. Hence, a suitable precondition for hybridization between *D. raddei* and *D. portshicnskii* does exist, but behavioural prezygotic isolation takes place as well. The absence of observations of interspecific mating support this statement. The ability of a lizard to distinguish between individuals of the same and different species is a well-known phenomenon. *Podarcis*, for instance, react differently to the scent of individuals of the same and different species (Gabirot et al. 2010). Rock lizards may also use visual signals (colour) at a distance for species discrimination. It is no surprise that visual and chemical signals allow lizards to distinguish between species, making them also capable of individual discrimination (Tsellarius and Men'shikov 1994; Lena and de Fraipont 1998; Bull 2000; Chapple and Keogh 2005) and establishing of personalised relationship between individuals in many species of reptiles including rock lizards (Tsellarius et al. 2016).

There are several possible explanations of why males do not mate with females of the other species. On the one hand, species recognition represents a fundamental mating threshold while on the other hand, sexual selection through mate choice may also determine mating success. It has been suggested that body size differences among related species promote reproductive isolation in vertebrates (Bolnick et al. 2006). Successful hybridization is possible between species if the size differences among males and female are minimal and Table 2Successful and failedintraspecies and interspecificmating attempts (N = 23) amongD. raddei and D. portschinskii.M, males; F, females

М	F	Mating attempts	Successful mating	Percentage success
D. portschinskii	D. raddei	0	0	0
D. portschinskii	D. portschinskii	6	5	83.3
D. raddei	D. raddei	15	9	60.0
D. raddei	D. portschinskii	2	0	0

comparable with those for conspecifics (Jančúchová-Lásková et al. 2015). This is the case for the studied species, where females have no significant differences between species in body size (50–60 mm). Hence, we have no reason to suggest body size influences the probability of hybridization between *D. raddei* and *D. portschinskii*.

Morphophysiological features are not the only mechanisms for interspecific isolation, which is often also based on differences in behaviour (Capula 1993). Male *D. raddei* and *D. portschinskii* differed in the way they restrain a female during copulation and its duration, which was longer in *D. portschinskii*.

Differences in reproductive cycles have been suggested to be the main isolation mechanism in the area of the secondary contact between *D. valentini* which reproduces in mid-May– early June and *D. raddei nairensis*, which reproduces in mid-June–July (Danielyan 1965). However, this might only partially explain the reproductive isolation between *D. raddei* and *D. portschinskii*, because this reproductive period in both species overlapped broadly in late May and early June, when males of both species were sexually active. Differences in belly coloration are the most striking external features of the studied species: the throat and belly in *D. raddei* is always green-yellow. As we know, species recognition in some species of lizards is based on coloration (Losos 2009). However, prezygotic isolation exists even among those species, which have similar coloration pattern of dewlap in *Anolis* (Ng et al. 2017).

Behav Ecol Sociobiol

(2019) 73:46

Morphological and physiological differences between related species are not the only features affecting social and sexual behaviour among the heterospecific individuals. Population density and sex ratio (Emlen and Oring 1977); ecological and social factors (Rosenthal 2013) affect sexual selection and mate choice. Hybridization often occurs in disturbed habitats, where the stable social structure is disrupted (Anderson, 1948). Proximal mechanisms of interbreeding include mate choice, which is based on individual preferences and environmental conditions (Rosenthal, 2013). Dreiss et al. (2010) reported weaker intraspecies sexual selection of males in male-biased populations of *Zootoca vivipara*. Vertebrates accept heterospecific partners if conspecific mates are rare or not available (Willis et al. 2011, 2012). The probability of interaction with the intraspecific sexual partner is

Fig. 5 Mating pattern in *D. portschinskii* (**a**) and *D. raddei* (**b**) and the jaw prints on the female's body after mating for *D. portschinskii* (**c**, on the belly) and *D. raddei* (**d**, on the femur) and a female of *D. raddei* with both types of print (**e**)



significantly reduced, if the sex ratio is biased. It may lead to the less selective mate choice and breeding with individuals belonging to different species (Dreiss et al. 2010). This suggestion is partly supported by the observations of populations comprising parthenogenetic and bisexual rock lizards. Parthenogenetic lizards commonly breed with bisexual species and produce backcrosses (Darevsky et al. 1985; Jančúchová-Lásková et al. 2015), if their number exceeds those of females belonging to bisexual species (Danielyan et al. 2008). For the studied species, this could have occurred under glacial conditions, when different parthenogenetic species appeared (Kearney 2005). The unstable climate and narrowing of the natural habitat areas potentially mixed initially isolated species, what could relax mate preference and interspecific breeding.

Acknowledgments Many thanks to Alexey Tsellarius, Natella Mirzoyan and two reviewers for their help in the improvement of the present manuscript and valuable critics of the text. We are grateful to Martin Whiting for his valuable suggestions and editing of the present manuscript.

Funding This work was supported by research projects RFBR 17-00-0425, RFBR 17-00-0430 and RFBR 18-54-05020.

Compliance with ethical standards

Ethical approval The research was approved by the Armenian Ministry of Nature Protection (N5/22.1/51043). All applicable international, national and institutional (Yerevan State and Moscow State Universities) guidelines for the care and use of animals were followed.

Conflict of interest The authors declare that they have no conflicts of interest.

References

Anderson E (1948) Hybridization of the habitat. Evolution 2:1-9

- Arakelyan MS (2001) Skeletochronological study of rock lizards from Armenia and some questions of their ecology. PhD thesis, Zoological Institute, St. Petersburg, Russia
- Arakelyan MS (2012) The speciation of bisexual species in sympatric populations of rock lizards of genus *Darevskia*. Biol J Armenia 64:10–15
- Arakelyan M, Danielyan F (2014) New evidence on hybrid zones of bisexual species of rock lizards of genus *Darevskia*. In: Proceedings of the International conference "Biological diversity and conservation problems of the fauna of the Caucasus-2", volume 2. Yerevan
- Arakelyan MS, Danielyan FD, Corti C, Sindaco R, Leviton AE (2011) Herpetofauna of Armenia and Nagorno-Karabakh. Society for the Study of Amphibians and Reptiles, Ithaca
- Arribas OJ (1999) Phylogeny and relationships of the mountain lizards of Europe and Near East (Archaeolacerta Mertrns, 1921, sensu lato) and their relationships among the Eurasian lacertid radiation. Rus J Herpetol 1:1–22
- Axtell RW (1972) Hybridization between western collared lizards with a proposed taxonomic rearrangement. Copeia 4:707–727
- Bolnick DI, Near TJ, Wainwright PC (2006) Body size divergence promotes postzygotic reproductive isolation in centrarchids. Evol Ecol Res 8:903–913

- Borkin LJ, Darevsky IS (1980) Reticular (hybridogeneous) speciation in vertebrate. Zh Obshch Biol 16:485–507 [In Russian]
 - Bull CM (2000) Monogamy in lizards. Behav Process 51:7-20
 - Capula M (1993) Natural hybridization in *Podarcis sicula* and *P. wagleriana* (Reptilia, Lacertidae). Biochem Syst Ecol 21:373–380
- Chapple DG, Keogh JS (2005) Complex mating system and dispersal patterns in a social lizard, *Egernia whitii*. Mol Ecol 14:1215–1227
- Cole CJ, Townsend CR (1990) Parthenogenetic lizards as vertebrate systems. J Exp Zool 256:174–176
- Cole CJ, Dessauer HC, Barrowclough GF (1988) Hybrid origin of a unisexual species of whiptail lizard, *Cnemidophorus neomexicanus*, in western North America: new evidence and a review. Am Mus Novit 2905:1–38
- Cole CJ, Taylor HL, Baumann DP, Baumann P (2014) Neaves' whiptail lizard: the first known tetraploid parthenogenetic tetrapod (Reptilia: Squamata: Teiidae). Breviora 539:1–19
- Csardi G, Nepusz T (2006) The igraph software package for complex network research. InterJournal, Complex Systems 1695. http:// igraph.org. Accessed 15 Oct 2018
- Cuellar O (1974) On the origin of parthenogenesis in vertebrates. Am Nat 963:625–648
- Danielyan FD (1965) Mechanisms of reproductive isolation in some Armenian forms of rock lizards (*Lacerta saxicola* Eversmann). Izvestia Akad Nauk Arm SSR (Biol) 18:75–80 [In Russian]
- Danielyan F, Arakelyan M, Stepanyan I (2008) Hybrids of *Darevskia* valentini, D. armeniaca and D. unisexualis from a sympatric population in Armenia. Amphibia-Reptilia 29:487–504
- Darevsky IS (1958) Natural parthenogenesis in certain subspecies of rock lizards. Dokladi Akad Nauk SSSR 122:730–732 [In Russian]
- Darevsky IS (1967) Rock lizards of the Caucasus (systematics, ecology, and phylogeny of the polymorphic lizards of the Caucasus of the subgenus Archaeolacerta). Nauka, Saint Petersburg Leningrad, published for the Smithsonian Institution and the National Science Foundation, Washington DC by the Indian National Scientific Documentation Centre, New Dehli
- Darevsky IS (1995) Epistandart evolution and hybridogenous speciation in reptiles. J Obshei Biol 56:310–315
- Darevsky IS, Kupriyanova LA, Uzzel T (1985) Parthenogenesis in reptiles. In: Gans C, Billet F (eds) Biology of the reptilian, vol 15. John Wiley and Sons Inc., New York, pp 412–526
- Dehghani A, Hosseinian S, Rastegar-Pouyani N, Mohammadpour A (2014) Sexual size dimorphism in *Darevskia raddei* (Sauria: Lacertidae) from Northwest Iran. Zool Middle East 60:120–124
- Dessauer HC, Cole CJ, Townsend CR (2000) Hybridization among western whiptail lizards (*Cnemidophorus tigris*) in southwestern New Mexico: population genetics, morphology, and ecology in three contact zones. Bull Am Mus Nat Hist 246:1–148
- Dreiss AN, Cote J, Richard M, Federici P, Clobert J (2010) Age- and sexspecific response to population density and sex ratio. Behav Ecol 21: 356–364
- Emlen ST, Oring LW (1977) Ecology, sexual selection, and the evolution of mating systems. Science 4300:215–223
- Fisher JW, Muth A (1989) A technique for permanently marking lizards. Herpetol Rev 20:45–46
- Freitas S, Rocha S, Campos J et al (2016) Parthenogenesis through the ice ages: a biogeographic analysis of Caucasian rock lizards (genus Darevskia). Mol Phylogenet Evol 102:117–127
- Fu J, MacCulloch R, Murphy R, Darevsky I (2000a) Clonal variation in the Caucasian rock lizard *Lacerta armeniaca* and its origin. Amphibia-Reptilia 21:83–89
- Fu J, MacCulloch R, Murphy R, Darevsky I (2000b) Divergence of the cytochrome b gene in the *Lacerta raddei* complex and its parthenogenetic daughter species: evidence for recent multiple origins. Copeia 2:432–440

- Gabirot M, Castilla AM, López P, Martín J (2010) Chemosensory species recognition may reduce the frequency of hybridization between native and introduced lizards. Can J Zool 88:73–80
- Galoyan EA (2013) Joint space use in a parthenogenetic Armenian rock lizard (*Darevskia armeniaca*) suggests weak competition among monoclonal females. J Herpetol 47:97–104
- Galoyan EA (2017) Unstable social structure indicates low diversity of relationships in the forest spotted skink *Sphenomorphus maculatus*. Amphibia-Reptilia 38:381–393
- Galoyan E, Moskalenko V, Gabelaia M, Tarkhnishvili D, Spangenberg V, Kolomiets O, Arakelyan M (2019) Why a new parthenogenetic rock lizard doesn't appear now: morphological, genetic and cytogenetic features of the two coexisting rock lizards *Darevskia raddei* and *D. portschinskii*. Amphibia-Reptilia (published online). https://doi. org/10.13140/RG.2.2.11844.48006
- Grismer JL, Grismer LL (2010) Who's your mommy? Identifying maternal ancestors of asexual species of *Leiolepis* Cuvier, 1829 and the description of a new endemic species of asexual *Leiolepis* Cuvier, 1829 from southern Vietnam. Zootaxa 2433:47–61
- Grismer JL, Bauer AM, Grismer LL, Jackmann T (2014) Multiple origins of parthenogenesis, and a revised species phylogeny for the Southeast Asian butterfly lizards, *Leiolepis*. Biol J Linn Soc 113: 1080–1093
- Husak JF, Fox SF (2003) Adult male collared lizards, *Crotaphytus collaris*, increase aggression towards displaced neighbours. Anim Behav 65:391–396
- Jančúchová-Lásková J, Landová E, Frynta D (2015) Are genetically distinct lizard species able to hybridize? A review. Curr Zool 61:155– 180
- Jenssen TA (1977) Morphological, behavioural and electrophoretic evidence of hybridization between the lizards, *Anolis grahami* and *Anolis lineatopus neckeri*, on Jamaica. Copeia 1977:270–276
- Kearney M (2005) Hybridization, glaciation and geographical parthenogenesis. Trends Ecol Evol 20:495–502
- Lena JP, de Fraipont M (1998) Kin recognition in the common lizard. Behav Ecol Sociobiol 42:341–347
- Losos JB (2009) Lizards in an evolutionary tree: ecology and adaptive radiation of anoles. University of California Press, Berkeley
- Moritz C, Uzzell T, Spolsky C, Hotz H, Darevsky I, Kupriyanova L, Danielyan F (1992) The material ancestry and approximate age of parthenogenetic species of Caucasian rock lizards (*Lacerta*: Lacertidae). Genetica 87:53–62
- Murphy RW, Fu J, MacCulloch RD, Darevsky IS, Kupriyanova LA (2000) A fine line between sex and unisexuality: the phylogenetic constraints on parthenogenesis in lacertid lizards. Zool J Linnean Soc 130:527–549
- Ng J, Geneva AJ, Noll S, Glor RE (2017) Signals and speciation: *Anolis* dewlap color as a reproductive barrier. J Herpetol 51:437–447
- Nicholson KE, Richards PM (2011) Home-range size and overlap within an introduced population of the Cuban Knight Anole, *Anolis equestris* (Squamata: Iguanidae). Phyllomedusa 10:65–73
- Panov EN, Zykova LY (2016) Rock lizards of Eurasia. KMK, Moscow
- Pianka ER, Vitt LJ (2003) Lizards. Windows to the evolution of diversity. University of California Press, Berkeley
- Pinho C, Kaliontzopoulou A, Carretero MA, Harris DJ, Ferrand N (2009) Genetic admixture between the Iberian endemic lizards *Podarcis bocagei* and *Podarcis carbonelli*: evidence for limited natural hybridization and a bimodal hybrid zone. J Zool Syst Evol Res 47: 368–377

- Rosenthal GG (2013) Individual mating decisions and hybridization. J Evol Biol 26:252–255
- Ruby DE (1978) Seasonal changes in territorial behaviour of the iguanid lizard *Sceloporus jarrovi*. Copeia 3:430–438
- Ryskov AP, Osipov FA, Omelchenko AV, Semyenova SK, Girnyk AE, Kochagin VI, Vergun AA, Murphy RW (2017) The origin of multiple clones in the parthenogenetic lizard species *Darevskia rostombekowi*. PLoS One 12:e0185161
- Schultz RJ (1973) Unisexual fish: laboratory synthesis of a "species". Science 179:180–181
- Spangenberg V, Arakelyan M, Galoyan E, Matveevsky S, Petrosyan R, Bogdanov Y, Danielyan F, Kolomiets O (2017) Reticulate evolution of the rock lizards: meiotic chromosome dynamics and spermatogenesis in diploid and triploid males of the genus *Darevskia*. Genes 8:149
- Tarroso P, Pereira RJ, Martínez-Freiría F, Godinho R, Brito JC (2014) Hybridization at an ecotone: ecological and genetic barriers between three Iberian vipers. Mol Ecol 23:1108–1123
- Taylor HL, Cole CJ, Hardy LM, Dessauer HC, Townsend CR, Walker JM, Cordes JE (2001) Natural hybridization between the teiid lizards *Cnemidophorus tesselatus* (parthenogenetic) and *C. tigris marmoratus* (bisexual): assessment of evolutionary alternatives. Am Mus Novit 3345:1–64
- Tsellarius AY, Men'shikov YG (1994) Indirect communications and its role in the formation of social structure in *Varanus griseus*. Russ J Herpetol 2:121–132
- Tsellarius AY, Tsellarius EY (2001) Dynamics of population spatial structure in *Lacerta saxicola* in deciduous forests of Navagir mountain ridge. Russ J Zool 80:1–8 [In Russian]
- Tsellarius AY, Tsellarius EY, Galoyan EA (2008) Interrelations of adults and juveniles of rock lizards—*Darevskia brauneri* (Reptilia, Zauria) in Navagir Mountain Ridge. Sovr Gerp 8:170–186 [In Russian]
- Tsellarius AY, Tsellarius EY, Galoyan EA (2016) Social relationships between males and females in the rock lizard (*Darevskia brauneri*, Lacertidae): 1. Friendly monogyny in males and polyandry in females. Biol Bull 43:1077–1086
- Uzzel T, Darevsky I (1973) The relationships of *Lacerta portschinskii* and *Lacerta raddei* (Sauria, Lacertidae). Herpetologica 29:1–6
- Verbeek B (1972) Ethologische Untersuchungen an einigen europäischen Eidechsen. Bonn Zool Beitr 23:122–151
- Vrijenhoek RC (1989) Genetic and ecological constraints on the origins and establishment of unisexual vertebrates. In: Dawley R, Bogart J (eds) Evolution and ecology of unisexual vertebrates. NY State Museum, Albany, pp 24–31
- Willis PM, Ryan MJ, Rosenthal GG (2011) Encounter rates with conspecific males influence female mate choice in a naturally hybridizing fish. Behav Ecol 22:1234–1240
- Willis PM, Rosenthal GG, Ryan MJ (2012) An indirect cue of predation risk counteracts female preference for conspecifics in a naturally hybridizing fish *Xiphophorus birchmanni*. PLoS One 7:e34802
- Wirtz P (1999) Mother species–father species: unidirectional hybridization in animals with female choice. Anim Behav 58:1–12

Publisher's note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.